Author’s response to reviews

Title: Risk Factors Affecting Prognosis in Metachronous Liver Metastases from WHO Classification G1 and G2 Gastroenteropancreatic Neuroendocrine Tumors after Initial R0 Surgical Resection

Authors:

Yang Lv (pwlvyang2015@163.com)
Xu Han (han.xu1@zs-hospital.sh.cn)
Xuefeng Xu (xu.xuefeng@zs-hospital.sh.cn)
Yuan Ji (ji.yuan@zs-hospital.sh.cn)
Yuhong Zhou (zhou.yuhong@zs-hospital.sh.cn)
Huichuan Sun (sun.huichuan@zs-hospital.sh.cn)
Jian Zhou (zhou.jian@zs-hospital.sh.cn)
Jia Fan (fan.jia@zs-hospital.sh.cn)
Wenhui Lou (lou.wenhui@zs-hospital.sh.cn)
Cheng Huang (huang.cheng@zs-hospital.sh.cn)

Version: 2 Date: 27 Feb 2019

Author’s response to reviews:

Dear Editor,

Thank you very much for your comments to our manuscript "Risk Factors Affecting Prognosis in Metachronous Liver Metastases from WHO Classification G1 and G2 Gastroenteropancreatic Neuroendocrine Tumors after Initial R0 Surgical Resection", of which the previous submission code is BCAN-D-18-03462R1.

In this letter, we will response to your kind technical comments:
A point-by-point response letter is as follows:

Technical Comments:

1. In your title page, please clearly indicate who the corresponding author of your paper is.

Response: Thanks for your comments. Title page is revised. We have two corresponding authors: Pro. Cheng Huang and Pro. WenHui Lou.

2. Please represent authors’ names using their full initials, not their full name, in the Authors’ Contributions section. If there are any duplicated initials, please differentiate them to make it clear that the initials refer to separate authors.

Response: We have already revised it. Thank you very much.

3. In the Funding section, please also describe the role of the funding body in the design of the study and collection, analysis, and interpretation of data and in writing the manuscript.

Response: We have added the role of funding body in the Funding Section. Thank you very much.

4. In your “ethical approval and consent to participate” section of your declarations please confirm whether written or verbal consent was obtained from participants. If written consent was not obtained, please explain why not.

Response: We confirmed that written consent was obtained from participants. Thank you very much.

5. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

Response: Accompanied with this letter, we upload a new manuscript. Thank you very much.
We hope our present version can address your kind and valuable comments. If you have any further comments and reviews, please do not hesitate to contact us.

Thank you very much

Best regards,

Cheng Huang and Wenhui Lou

Zhongshan Hospital, Fudan University, Shanghai China